Author's response to reviews

Title: Axial Torsion of Meckel's Diverticulum. A Rare Complication of a Common Variant: a case report

Authors:

AJAI SETH (dr.ajaiseth@gmail.com)
JAI SETH (seth.jai@gmail.com)

Version: 2 Date: 9 January 2011

Author's response to reviews: see over
Dear Editor

Thank you for your kind comments. In line with your suggestions we have made the following changes.

- The title has now been changed to your suggested title.
- With regards to preoperative investigations, we have stated the relevant blood test results. It is important to note that the patient only had an x-ray (no CT or US) as a clinical diagnosis of appendicitis was made and the patient taken straight to theatre. This has now been made clear.
- The patient’s recovery was unremarkable and his discharge and follow-up has been stated.
- The discussion has been reorganised with the headings removed and the aetiology section moved to the beginning as suggested.
- The conclusion section has now been created with the addition of your suggested sentence.
- The references you mentioned have been incorporated into the manuscript.
- The image has been changed to a clearer intra-operative picture as per the referee’s suggestion.

The main changes in the manuscript have been tracked in red for your convenience. I hope the manuscript is now suitable for publication. If there is anything else that requires attention, please let me know.

Yours sincerely

Dr Ajai Seth

dr.ajaiseth@gmail.com